Author's response to reviews

Title: Trans-subclavian Approach for Radiofrequency Ablation of Premature Ventricular Contractions originating from subtricuspid annulus

Authors:

Li teng (gdmc99@163.com)
Xian-zhang Zhang (gdzxz@163.com)
Ping-zhen Yang (ypz@163.com)
Yu-mei Xue (xymgdc@163.com)
Xian-hong Fang (fangxianhong2000@yahoo.com)
Hong-tao Liao (hometown1975@163.com)
shulin wu (wushulin8888@yahoo.com.cn)

Version: 2 Date: 21 August 2012

Author's response to reviews: see over
Cover Letter:

1. Warrants that this manuscript of Trans-subclavian Approach for Radiofrequency Ablation of Premature Ventricular Contractions originating from subtricuspid annulus has been reviewed and approved by all named authors.

2. I is empowered by all of the authors to act on their behalf with respect to the submission of the manuscript.

3. Confirm that the manuscript is original.

4. Confirm that the manuscript does not infringe upon any copyright or other proprietary right of any third party.

5. Confirm that neither the text nor the data reported have been published previously.

6. The Ethics Committee of the Guangdong Cardiovascular Institute approved this case.

7. The article or a substantially similar article is not under consideration by another journal at this time.

8. None of the authors has any potential financial conflict of interest related to this manuscript.